Author’s response to reviews

Title: A CROSS SECTIONAL STUDY ON NURSING PROCESS IMPLEMENTATION AND ASSOCIATED FACTORS AMONG NURSES WORKING IN SELECTED HOSPITALS OF CENTRAL AND NORTHWEST ZONES, TIGRAY REGION, ETHIOPIA

Authors:

Zeray Baraki (zer2mun.dan@gmail.com)
Fiseha Girmay (fiseha1999@yahoo.com)
Kalayou kidanu (kalushaibex@gmail.com)
Hadgu Gerensea (hadeguellen1@gmail.com)
Dejen Gezehegn (dejen.8000@gmail.com)
Hafte Teklay (haftishkind@gmail.com)
Sue Anne Bell (sabell@umich.edu)

Version: 1 Date: 12 Jan 2017

Author’s response to reviews:

To
The Editor-in-Chief,
BMC Nursing Research Journal

Sub: Submission of original revised manuscript

Dear Editor:

With reference to the above, please find my submission of revised manuscript entitled with “a cross sectional study on nursing process implementation and associated factors among nurses working in selected hospitals of Central and Northwest zones, Tigray region, Ethiopia” for possible publications with a point-by-point response to the comments of reviewer as follows.
Point by point response for each raised comment

1. Epi Info is a data entry software and the right spelling is “Epi Info”. Accordingly I have made it the right one.

2. In Abstract result section, I have made the amendment according the given comment

3. Page 2 line 5 I have made “Theses greatly affect” to “These greatly affect”

4. I have made the amendment according the comment on Page 6, 7, 11, 13 and 15

5. I have made lower case the reference 16

6. Additionally I have made BMC nursing reference style in the End note

NB: - All the amendments are in green color in the manuscript.